Author’s response to reviews

Title: Medical and moral considerations regarding complex medical decisions in older patients with multimorbidity: a compact deliberation framework

Authors:

Jeroen F.A.M. Janssens (j.f.a.m.janssens@lumc.nl)

Susanne J. de Kort (susannedekort@hotmail.com)

Wilco P. Achterberg (w.p.achterberg@lumc.nl)

Susan Kurrle (kurrle@bigpond.com)

Ngaire Kerse (n.kerse@auckland.ac.nz)

Ian D. Cameron (ian.cameron@sydney.edu.au)

Dorothea P. Touwen (d.p.touwen@lumc.nl)

Version: 5 Date: 21 Dec 2017

Author’s response to reviews:

In the manuscript your remarks concerning:

- Cameron being part of the editorial board
- using initials in stead of full names,
- using the headings Background, Main text, Conclusions have been covered

A statement regarding consent to publish this information from the participants next of kin is uploaded under 'attached files'.